\begin{table}%T3
\caption{\label{tab:seyferts}All Seyferts observed with VLBI. Column~1 lists the names we
used, Col.~2 the names used by \cite{Veron2001}, Col.~3 the
Seyfert types as described by \cite{Veron2001}, Col.~4 the linear
resolution in~pc, Col.~5 the instrument and frequency used, and~Col.~6 the reference to the publication.}
%\centering
\par
\begin{tabular}{lllrll}
\hline
\hline\noalign{\smallskip}
\multicolumn{1}{c}{Name} 
& \multicolumn{1}{c}{V\'eron name} 
& \multicolumn{1}{c}{Class}
& \multicolumn{1}{c}{Lin. Res.}
& \multicolumn{1}{c}{Instrument}
& \multicolumn{1}{c}{Ref.}
\\
&
&
& \multicolumn{1}{c}{(pc)}\\
\hline					\object{3C 287.1}    &                 & S1   & 5.43      & VLBA 5 GHz                    & \cite{2000ApJS..131...95F}\\  
\object{3C 390.3}    &                 & S1.5 & 1.09      & global VLBI 5 GHz             & \cite{1996AaA...308..376A}\\  
\object{III Zw 2}    &                 & S1.2 & 0.260     & VLBA 15/43 GHz                & \cite{2000AaA...357L..45B}\\  
\object{Ark 564}     &                 & --    & 0.598     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{IC 5063}     & PKS 2048-57     & S1h  & 3.30      & LBA 2.3 GHz + LBA HI          & \cite{2000AJ....119.2085O}\\  
\object{MCG 8-11-11} &                 & S1.5 & 0.635     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{Mrk 1}       &                 & S2   & 6.18      & EVN  1.7 GHz                  & \cite{1999ApJ...518..117K}\\  
\object{Mrk 1}       &                 & S2   & 0.550     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{Mrk 3}       &                 & S1h  & 5.24      & EVN  1.7 GHz                  & \cite{1999ApJ...518..117K}\\  
\object{Mrk 78}      &                 & S2   & 1.75      & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{Mrk 231}     &                 & S1.0 & 0.343     & VLBA 15 GHz                   & \cite{1999ApJ...517L..81U}\\  
\object{Mrk 231}     &                 & S1.0 & 4.09      & VLBA 1.4/2.3/4.8/8.4/15/22 GHz& \cite{1999ApJ...516..127U}\\  
\object{Mrk 273}     &                 & S2   & 7.32      & VLBA 1.4 GHz                  & \cite{2000ApJ...532L..95C}\\  
\object{Mrk 348}     &                 & S1h  & 0.131     & VLBA 15 GHz                   & \cite{1999ApJ...517L..81U}\\  
\object{Mrk 348}     &                 & S1h  & 6.41      & EVN 1.4 GHz                   & \cite{1983AaA...128..318N}\\  
\object{Mrk 348}     &                 & S1h  & 1.49      & VLBA 1.7/5.0/8/15/22 GHz      & \cite{2003ApJ...590..149P}\\  
\object{Mrk 477}     &                 & S1h  & 2.39      & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{Mrk 530}     & NGC 7603        & S1.5 & 1.45      & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{Mrk 766}     &                 & S1.5 & 0.413     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{Mrk 926}     &                 & S1.5 & 0.917     & VLBA 8.4 GHz                  & \cite{2000ApJ...529..816M}\\  
\object{Mrk 1210}    &                 & S1h  & 0.392     & EVN 1.6/5 GHz VLBA 1.6/5 GHz  & This paper                \\  
\object{Mrk 1218}    &                 & S1.8 & 0.582     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{NGC 1052}    &                 & S3h  & 0.015     & VLBA 5/8.4/22/43 GHz          & \cite{Kadler2002}         \\  
\object{NGC 1052}    &                 & S3h  & 0.048     & VLBI 1.4-43.2                 & \cite{2003AaA...401..113V}\\  
\object{NGC 1052}    &                 & S3h  & 0.038     & VLBA 2.3/8.4/15.4 GHZ         & \cite{2001PASJ...53..169K}\\  
\object{NGC 1068}    &                 & S1h  & 0.221     & VLBA 1.7,5,15 GHz             & \cite{1998ApJ...504..147R}\\  
\object{NGC 1068}    &                 & S1h  & 2.94      & EVN 1.4 GHz                   & \cite{1987AJ.....93...22U}\\  
\object{NGC 1167}    &                 & S3   & 0.895     & global VLBI 5 GHz             & \cite{2001ApJ...552..508G}\\  
\object{NGC 1167}    &                 & S3   & 0.639     & global VLBI 1.6 GHz           & \cite{1990ApJ...358..159G}\\  
\object{NGC 1275}    &                 & S1.5 & 0.416     & VLBA 2.3/5/8.4/15.4/22/43 GHz & \cite{2000ApJ...530..233W}\\  
\object{NGC 2110}    &                 & S1i  & 0.151     & VLBA 8.4 GHz                  & \cite{2000ApJ...529..816M}\\  
\object{NGC 2110}    &                 & S1i  & 0.966     & EVN 1.6/5 GHz                 & This paper                \\  
\object{NGC 2273}    &                 &  --   & 0.342     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{NGC 2639}    &                 & S3   & 0.233     & global VLBI 5 GHz             & \cite{VirLal2001}         \\  
\object{NGC 2639}    &                 & S3   & 0.162     & VLBA 1.6/5/15 GHz             & \cite{1998ApJ...505..587W}\\  
\object{NGC 3079}    &                 & S2   & 0.044     & VLBA 5/8/22 GHz               & \cite{1998ApJ...495..740T}\\  
\object{NGC 3079}    &                 & S2   & 0.054     & global VLBI, 5.0 GHz          & \cite{1988ApJ...335..658I}\\  
\object{NGC 3079}    &                 & S2   & 0.042     & global VLBI 1.4/8.4/15/22     & \cite{2000PASJ...52..421S}\\  
\object{NGC 3147}    &                 & S2   & 0.219     & VLBA 1.6/2.3/5/8.4 GHz        & \cite{2001ApJ...562L.133U}\\  
\object{NGC 3227}    &                 & S1.5 & 3.74      & MERLIN 1.6/5 GHz              & \cite{1995MNRAS.275...67M}\\  
\object{NGC 4151}    &                 & S1.5 & 0.116     & VLBA 1.6,5 GHz                & \cite{1998ApJ...496..196U}\\  
\object{NGC 4151}    &                 & S1.5 & 1.29      & EVN 1.7 GHz                   & \cite{1986MNRAS.218..775H}\\  
\object{NGC 4151}    &                 & S1.5 & 4.82      & MERLIN 5 GHz                  & \cite{1993MNRAS.263..471P}\\  
\object{NGC 4168}    &                 & S1.9 & 0.236     & VLBA 5 GHz                    & \cite{2002AaA...392...53N}\\  
\object{NGC 4203}    &                 & S3b  & 0.084     & VLBA 1.6/2.3/5/8.4 GHz        & \cite{2001ApJ...562L.133U}\\  
\object{NGC 4258}    &                 & S2   & 0.029     & VLBA 22 GHz                   & \cite{1997ApJ...475L..17H}\\  
\object{NGC 4258}    &                 & S2   & 0.232     & VLBA 1.4/1.6 GHz              & \cite{2000ApJ...536..675C}\\  
\object{NGC 4395}    &                 & S1.8 & 0.221     & VLBA 1.4 GHz                  & \cite{2001ApJ...553L..23W}\\  
\object{NGC 4565}    &                 & S1.9 & 0.207     & VLBA 5 GHz                    & \cite{2000ApJ...542..197F}\\  
\object{NGC 4579}    &                 & S3b  & 0.118     & VLBA 1.6/2.3/5/8.4 GHz        & \cite{2001ApJ...562L.133U}\\  
\object{NGC 4579}    &                 & S3b  & 0.246     & VLBA 5 GHz                    & \cite{2000ApJ...542..197F}\\  
\object{NGC 5252}    &                 & S2   & 0.671     & VLBA 8.4 GHz                  & \cite{2000ApJ...529..816M}\\  
\object{NGC 5506}    &                 & S1i  & 0.815     & EVN 1.6/5 GHz                 &      This paper           \\  
\object{NGC 5548}    &                 & S1.5 & 0.366     & VLBA 8.4 GHz                  & \cite{2000ApJ...531..716W}\\  
\object{NGC 5793}    &                 & S2   & 0.135     & VLBA 1.7,8.4,15,22 GHz        & \cite{2001ApJ...560..119H}\\  
\object{NGC 5793}    &                 & S2   & 0.316     & VLBA+Y27 1.6/5 GHz            & \cite{2000AaA...360...49H}\\  
\hline
\end{tabular}
\end{table}